Author’s response to reviews

Title: Transcatheter Occlusion of Giant Congenital Coronary Cameral Fistulae: Case reports

Authors:

Deogratias Arnold Nkya (gratias1980@gmail.com)
Greenwood Sinyangwe (gsinyangwe@gmail.com)
Farirai Takawira (fari.takawira@up.ac.za)

Version: 2 Date: 31 Jul 2019

Author’s response to reviews:

Past medical history and family history for case 1, 2 and 3 are added on pages 3, 5, and 6 respectively.

Case 1: There is no past history of recurrent illness or hospital admission. There is no history of congenital heart diseases in the family.

Case 2: There was no significant past medical history and he was not receiving any chronic medication.

Case 3: She never smoked or consumed alcohol.

Language editing of the whole document was performed by a professional editor. The editing certification document is attached as a supplementary material.

All changes and editing are marked in colour and track changes on the revised document.